Author’s response to reviews

Title: The effects of occupational pollutants on the reproductive health of female informal street traders in Warwick Junction, Durban, South Africa – a cross-sectional study

Authors:
Sujatha Hariparsad (HariparsadS1@ukzn.ac.za)
Rajen Naidoo (NaidooN@ukzn.ac.za)

Version: 3 Date: 19 Nov 2019

Author’s response to reviews:

TECHNICAL COMMENTS RESPONSE

Authors Contributions: the statement has been included under Authors’ Contributions to stipulate the role of both authors in reviewing manuscript.

Funding: the role of funders has been defined.

Figure: the figure has been removed as it does not add value to the manuscript in terms of specific knowledge regarding the outcomes and exposures of the sample.

Supplement: the oral conference abstract has been referenced on “Discussion Line 3” as reference 20.

Reviewer’s comments: these have been addressed regarding use of the word “causality” the word has been removed from Abstract and Background